

Review



# Towards 100 Gbps over 100 km: System Design and **Demonstration of E-Band Millimeter Wave Communication**

Zeyuan Zhang, Xianbing Zou, Qun Li and Ning Wei \*

National Key Laboratory of Science and Technology on Communications, University of Electronic Science and Technology of China (UESTC), Chengdu 611731, China

\* Correspondence: wn@uestc.edu.cn

Abstract: Long-range E-band communication with fiber-equivalent speed is emerging extensively as a critical technology in the next-generation communication. This paper firstly reviews the relevant progress in recent research. A brief survey is presented on high-speed, long-range E-band communication systems and their relevant techniques that are essential to the link design, including antenna, power amplifier (PA), channel, and digital baseband processing. In the second part, we review our recent field trial of a long-range air-to-ground E-band link, which maintains steady transmission from a slow-moving helium balloon to the ground station with a vertical dimension of 20 km. The improvement directions and future research topics are then discussed.

Keywords: E-band millimeter wave; LoS MIMO; air-to-ground; long-range wireless communication system



Citation: Zhang, Z.; Zou, X.; Li, Q.; Wei, N. Towards 100 Gbps over 100 km: System Design and Demonstration of E-Band Millimeter Wave Communication. Sensors 2022. 22,9514. https://doi.org/ 10.3390/s22239514

Academic Editor: Francisco Falcone

Received: 30 October 2022 Accepted: 29 November 2022 Published: 5 December 2022

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations



Copyright: © 2022 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/).

## 1. Introduction

Along with the increasing bit rate in modern wireless access and backhaul transmission, the communication requirements in exceptional cases such as disaster recovery also underlie the growing demand for wireless systems with long range and high capacity. The exploitable bandwidth at high millimeter-wave frequencies has become an effective choice to meet the requirement. However, due to the solid atmospheric absorption, the communication range is severely limited for frequency bands around 60 GHz, 120 GHz, and 180 GHz, typically less than 20 km [1]. With 10 GHz available bandwidth right in the atmospheric window and robust transmission ability through dust and fog, the E-band (71–76 GHz, 81–86 GHz) communication systems are considered promising candidates to achieve 100 Gbps data rate at a range over 100 km, which is regarded as the technological objective in DARPA 100 Gb/s RF Backbone funding opportunity [2].

Given that signals in higher frequency have poor diffraction ability, line of sight (LoS) propagation in the millimeter wave is primarily considered under most validated scenarios. A terrestrial E-band link could be more easily impaired by shadow fading and earth curvature compared to other cases, such as aeronautics and air-to-ground communication. However, with the advance in antenna, high-power semiconductors, and advanced DSP techniques, various E-band communication systems with extensive coverage (tens of km) and high capacity (tens of Gbps) have been extensively emerging, both in terrestrial [3–7], and air-to-ground [8,9] scenario.

In this paper, we discuss in detail the system design and implementation of highcapacity, long-range E-band link. Thereafter, to further investigate the in-depth characteristic of the E-band air-to-ground channel and verify the platform's performance in hand, a field transmission trial was also conducted.

The remainder of this paper is organized as follows. Section 2 introduces a brief survey of high-capacity, long-range E-band communication systems demonstrated in recent years and details the enabling technology essential to the link design. Section 3 reviews our field trial of an E-band air-to-ground link operated at 74 GHz from a slow-moving helium

balloon to the ground station with a vertical dimension of 20 km. Section 4 discusses the encountered bottleneck and potential improvement in the future. Section 5 concludes the work.

## 2. Review on System Design

## 2.1. System-Level Demonstration

A brief survey of high-speed, long-range E-band wireless communication links in recent years is presented in this part. Notably, all works have reached a transmission rate over tens of Gbps at a considerable distance (see Figure 1).



**Figure 1.** E-band long-range communication systems in the literature [3–9]. The two-way and one-way arrows represent the bidirectional and unidirectional link, respectively.

The authors in [3] demonstrated a 1.7 km wireless link based on 2 × 2 multiple input multiple output (MIMO). Photonic millimeter wave generation was used to generate 85.5 GHz carrier frequency, after QPSK modulation and separated by a polarization beam splitter, the signal was amplified and fed into parallel 45 dBi Cassegrain antennas with H- and V-polarization, respectively, namely single polarized MIMO. The experimental result showed under the bit-error rate of  $3.8 \times 10^{-3}$ , the link can realize up to 20 Gbps data delivery at 1.7 km.

The terrestrial link proposed in [4,5] achieved up to 6 Gbps QPSK signal transmission at 36.7 km and 15 Gbps 8-QAM signal transmission over 15 km without applying multiplexing scheme. The power amplifiers were fabricated in GaAs/GaN mHEMT and provide about 80 dBm equivalent isotropic radiated power (EIRP) along with the 50 cm parabolic antenna. The Tx and Rx antennas were optically aligned to find the maximum receive power, and an overhead of around 15 dB is preserved for the atmospheric loss variations. The result showed that system bit error rate (BER) under rain and fog ( $4.6 \times 10^{-3}$ ) is inferior to that under clear sky ( $6.6 \times 10^{-8}$ ). Under the assumption of an ideal additive white Gaussian noise (AWGN) channel, the signal-to-noise ratio (SNR) generally has a 5 dB deterioration in comparison to the theoretical value derived from the link budget. However, the measured atmospheric loss showed good consistency with the ITU model, with a deviation of around 0.2 dB.

Furthermore, the link proposed in [6] provides an overall 80 Gbps bi-directional data rate over a range of 16 km. The transmit channel (71–76 GHz) and receive channel (81–86 GHz) are each separated into two subchannels with 2 GHz bandwidth, and the transmit waveform is duplexed with a corresponding receive waveform using high isolation duplexer. Thereafter, the polarization multiplexing was conducted by using an orthomode transducer (OMT). By doing so, four of the 2 GHz subchannel each sank by a BSM85100 with 10 Gbps data rate was multiplexed at the transceiver. E-band GaN power amplifiers

with 1W saturated power were applied before each of the eight sub-channels combination to eliminate an inevitable 3 dB backoff. With 16-QAM Modulation, the link sustained an error-free (BER <  $1 \times 10^{-8}$ ) bi-directional data rate of 44 Gbps at an equivalent range of 30 km. By applying 128-QAM, the system can provide an overall bi-directional bit rate of 76.2 Gbps (65 Gbps error-free) over 16 km.

To further increase the spectrum efficiency, Ericsson has presented a 139 Gbps LoS-MIMO link at 1.5 km operated at lower E-band with 2.5 GHz channel bandwidth [7]. Four 51 dBi parabolic dual-polarized antennas were deployed at both transmitter and receiver to set up an  $8 \times 8$  LoS MIMO system. Limited by noise, the highest throughput was measured 139 Gbps with 256-QAM compared to the theoretical value of 144 Gbps (2.25 Gbaud×8 bits/symbol×8 streams), resulting in a spectrum efficiency of 55.6 bps/Hz. Moreover, the research has tested the robustness of both optimal and suboptimal deployments, i.e., offsetting the vertical separation between the receive antennas by 0.4 m and by 0.8 m.

In terms of air-to-ground transmission, the authors in [8] demonstrated a simplex communication link operating in 71–76 GHz between a microlight aircraft and a ground terminal. The aircraft flew with 1000 m height above the ground with a constant speed of 100 km/h, while the link can sustain a stable rate of 9.8 Gbps within 12 km. At RF frontend, GaN based solid-state power amplifier (SSPA) with 30 dBm output power was applied, and a 48.7 dBi steerable Cassegrain antenna at the ground terminal was aligned towards the aircraft based on GPS position information, whereas the transmit antenna with 39.7 dBi gain is packaged in a gimbal programmed to continuously point at the ground station to adaptively compensate the aircraft movement. Several modulation schemes were tested, the link BER is  $7 \times 10^{-4}$  when 8 PSK modulation is applied in a 9.6 Gbps data rate.

In [9], the authors presented test results of an E-band link between a fixed ground station and a Cessna aircraft that flew at a top speed of 463 km/h in a 12 km air-toground slanted path, the link has achieved a bidirectional peak data rate of 40 Gbps. The transceiver aggregated  $4 \times 10$  Gbps modulated carriers using a combination of frequency and polarization multiplexing. GaAs PAs with 33 dBm saturated output power was applied before signal was radiated by a 120 cm huge axially displaced ellipse (ADE) dual-reflector antenna with 59 dBi gain at ground terminal, and an 18 cm 42 dBi ADE at airborne. The highly sharp beamwidth was aligned by Global Position System (GPS)-based pointing, acquisition and tracking (PAT) process with accuracy better than 0.05°. Remarkably, the terminal provides the maximum bidirectional 40 Gbps data rates up to an altitude of 28 km and 10 Gbps up to 310 km.

#### 2.2. E-Band Antenna

The nonnegligible path attenuation and the atmospheric variation in long-range pointto-point communication links make signal power extremely precious for the systems. In E-band long-range communication, the essential requirement for antennas, except for the gain performance, is steerable beam, since antenna mast movement can increase the risk of outage by introducing additional attenuation (see Figure 2). This is further discussed in [10]. In this regard, the high-gain reflector/lens antennas with mechanical alignment and phased arrays seem to be the winning solution. Conventional Cassegrain antenna configuration suffers from secondary reflector blockage. Though the impairment can be compensated by shaping techniques and making a pointed vertex, it is still challenging to design an efficiently shaped Cassegrain reflector at higher frequency in the presence of diffraction effects [11]. ADE geometry can effectively address these problems with a displaced tilted ellipse along the axis, which eliminates the blockage due to the sub-reflector and the feed.



Figure 2. Antenna movement with degree of freedom 6.

In [12], an E-band ADE reflector antenna with adjustable beamwidth was designed, reaching the highest directivity of 46.1 dBi. The adjustable beamwidth was achieved by mechanically changing the focus distance, which is also known as zoom antenna architecture. The half-power beam width (HPBW) of the proposed antenna can be expanded from 0.88° to 4.4° in the lower E-band and from 0.86° to 5.6° in the upper E-band. This variation in beamwidth is of great value in beam switching and the PAT process, allowing frontends steering with more flexibility. For a wider scan range, ref. [13] presented an E-band integrated lens antenna (ILA) fed by a 64-element beam-switching patch array integrated with diode switches on a multilayered liquid crystal polymer (LCP) substrate. The beam-switching range of  $\pm 4^{\circ} \times \pm 17^{\circ}$  was achieved by changing the active antenna elements, which was controlled by an on-board switching network. The proposed ILA was applied later in an experimental E-band link to convey 700 Mbps data stream at a range of 55 m. However, the EIRP was constrained due to the significant mismatch of diode switches. The polarization multiplexing, enabled by polarized antennas, is commonly used in E-band antenna arrangement to improve the spectrum efficiency. Ref. [14] presented a high-gain ADE antenna fed by dual circular polarized horn antenna worked in W-band that reached directivity of 50 dBi. The cross-polarization discrimination (XPD) was higher than 28 dB for both left hand circular polarization (LHCP) and right hand circular polarization (RHCP). Another W-band dual-polarization reflector antenna work with RF sum-difference network was realized in [15]. The feeding network was composed of four OMTs and a coupling retracted structure. The sum of the opposite port is used to generate the sum beam of different polarization, while the difference in the opposite port is used to form the azimuth and pitch difference signal. By this means, the dual-polarization mono-pulse angle measurement can be conducted to identify the direction of arrival (DoA) of the incident wave, with a range of  $\pm 3^{\circ}$ . The proposed antenna is measured to have 41 dBi gain, with remarkable cross-polarization level of 43 dB.

When it comes to phased arrays, according to the publication to date, the application still stays rare in E-band long-range transmission. This is partly brought by the manufactural challenge encountered by the E-band phase array, that is, with the operation wavelength around 4 mm, the overall antenna architecture design faces problems such as space constraints in circuit integration, concentrated heat dissipation, poor scalability, and strict manufacturing tolerances. In [16], a 16-element highly integrated W-band phased array, including the transceiver chipset manufactured in 0.18-µm SiGe BiCMOS technology was presented. The transceiver system integrates converters, phase lock loop (PLL), a 16-channel beamforming network, and integrated low noise amplifier (LNA) and PA for the receiver and transmitter, respectively. The transceiver integrated circuit (IC) was flip-chipped onto a low-cost organic printed circuit board (PCB) with two differently polarized antenna arrays fed by coupling apertures and stacked via. The measured EIRP of

the transceiver is 34 dBm in each polarization which is enough to achieve 60 Gbps data transmission using 64-QAM at a distance of 1 m. The work was carried forward in [17] where the active phased array was extended into 384-Element. A maximum saturated EIRP of 60 dBm (1 kW) is measured at the boresight of the 256 transmit elements. The scalable architecture is preserved by incorporating the radio frequency integrated circuit (RFIC) and antennas into a PCB tile. The authors also established a wireless link operating at 90.7 GHz using the proposed phased array transceiver. The TX reached 52 dBm EIRP with an 8 dB backoff, 10 Gbps 16-QAM signal was demodulated with error vector magnitude (EVM) less than 8% over a distance of 265 m. In [18], a scalable 256-element E-Band active phased array was implemented on a 19-layer low temperature co-fired ceramic (LTCC) substrate, integrated with transceiver components fabricated in SiGe BiCMOS process. Due to the cost and system complexity constraints, hybrid beamforming was performed with 8 elements randomly tiled rectangular sub-arrays. The active phase array was tested in a 73 GHz link at 20 m with an EIRP of 35.5 dBm.

High-gain reflectors are mostly mechanically steered, which fails to accommodate the demand for fast networking and beam switching. Phased arrays can provide quick electronic scanning and flexible beam forming, but large phased arrays are more expensive and sometimes unpractical. The phased array fed reflector (PAFR), which received widespread attention in the radio astronomy community [19–22] may become an intermediate option to provide both considerable gain performance and a relatively limited electronic scanning region (usually 20°) and, of course, with lower cost. Generally, PAFR is composed of a reflector and a small phased array antenna on its focal plane, yet various forms of PAFR can be conducted by designing the reflector architectures (center/offset direct-fed, dual reflectors) and feed type (analog/digital phased arrays, switch matrix) [23].

The biggest challenge encountered by PAFR is the limitation of illumination area, that is, only limited array elements can be located in the area of maximum power density, which results in a limited EIRP level and the beam-scanning capability. As reported in [24], by increasing the axial displacement, a higher EIRP can be achieved in the PAFR antenna. For the mentioned 9 imes 9 array fed reflector system working at 71 GHz, an axial displacement of 2.3  $\lambda$  towards the reflector could provide an EIRP increase of 8 dB. However, this comes at the cost of a larger array size to meet the 80% efficiency criterion. Another challenge for conventional PAFR is the limited scanning range due to the beam deviation that occurs in the focal plane while performing wide-angle scanning. Ref. [25] introduced mathematical analysis to model the slightest deviation of the focused beam during wideangle scanning when different kinds of reflectors were applied. The parabolic reflector is verified to have the largest illuminated region, which enables more elements to be involved in reflection, while during the wide-angle scanning process, a spherical reflector without axial displacements is found to be the best option for minimizing the lateral array size since it has the slightest deviation from the axis of revolution. In [26], a double shaped-reflector for focal plane array with wide FoV is presented. The limitation of illumination area and scan FoV in conventional PAFRs is addressed by introducing a secondary polynomial surface to gain extra control of the waves interacting with the array surface. By optimizing the geometric configurations, the proposed PAFR system can achieve scan angles up to  $5^{\circ}$  with 90% of array elements involved. For a scan angle up to  $10^{\circ}$ , the whole array is active. This comes at the cost of design complexity and sufficient size of the sub-reflector. However, the result showed a significant enhancement of 15 dB in EIRP compared to an equivalent conventional PAFR. On the extreme wide scanning scheme, The Lockheed Martin Corporation has proposed a novel architecture of PAFR which is called Wide Angle ESA Fed Reflector (WAEFR) [27]. An annular field of regard is created using ring-focus optics in dual reflectors. By custom beamforming, the system eliminated the need for an enormous feed array in wide-angle scan procedure. A zenith-facing electronically scanned array (ESA) working in conical scan mode and an ESA feed dual-reflector can cover the entire hemispherical FoV.

#### 2.3. E-Band Power Amplifier

The PA plays a crucial role in the RF frontend of E-band long-range communication, for raising the transceiver EIRP could be the most straightforward solution to increase the SNR at the receiver. Consider a 73 GHz E-band link with 2 GHz bandwidth. Assume the required SNR at uncoded BER  $1 \times 10^{-6}$  (AWGN) is 12 dB, which is sufficient for QPSK demodulation. The noise figure of receiver *NF* is assumed to be 8 dB, and the antenna noise temperature  $T_a = 290$  K. So the system noise temperature of the receiver system can be obtained as

$$T_{svs} = T_a + (10^{NF/10} - 1)T_0 = 1830K$$
(1)

where  $T_0 = 290$  K, according to

$$SNR_o = \frac{S_i}{(kT_{SYS}B)}$$
(2)

We have the receiver input  $S_i = -61$  dBm. where  $k = 1.38 \times 10^{-23}$  J/K denotes Boltzmann constant and *B* denotes system bandwidth. For atmosphere attenuation of 0.4 dB/km, if the antenna gain is fixed at 40 dBi and a margin of 10 dB is preserved, a communication range of 20 km would require a PA with at least 33 dBm saturation power.

A survey of published PAs in different technologies is shown in Figure 3. Though the PAs fabricated in Si-based technologies are superior in price and integrability, it can be observed that they are apparently limited by the output power. The InP-based technology excels by outstanding power performance above 100 GHz. The GaAs and GaN-based processes are considered more suitable for RF frontend in E-band communication. GaAs process can provide exceptional power density with great flexibility, which enables its use in different scenarios. However, when it comes to a watt level performance above 70 GHz, GaN seems to be the only promising candidate [28].



**Figure 3.** A Survey of the published E-band PAs, the plot shows saturated output versus frequency for different technologies, according to the data from [29].

Under the funding of DARPA, the HRL lab has released a prototype monolithic microwave integrated circuit (MMIC) PA with 20 dB gain from 79–95 GHz using 40 nm GaN T3 process [30], the process can realize up to 200 GHz  $f_T$ , 400 GHz  $f_{MAX}$  and a breakdown voltage up to 40 V, which provides PA with higher linearity, dynamic range, and RF survivability. The proposed MMIC PA achieves an output power of 1.3 W (31.3 dBm) at a remarkable power-added efficiency (PAE) of 27%. An 8.3 mm × 2.5 mm four-stage power amplifier circuit is proposed in [31] using OMMIC's 60 nm GaN HEMT process, the saturated output power of the proposed PA is over 1.8 W (32 dBm), and the linear gain is

more than 20 dB. It is worth noting that the remarkable flatness of  $\pm 0.2$  dB is achieved to ease the burden of single carrier-frequency domain equalization (SC-FDE) systems. The PA operated in 71–76 GHz with PAE higher than 25%, which is also friendly to small size, weight, and power (SWaP) applications.

To further increase the output power, power combining using waveguide and low-loss transmission line media is widely applied to E-band PA to achieve 10 W+ output [32]. Ref [33] reported a W-band GaN power amplifier which could provide output power of 37 W (45.7 dBm) across the 75 to 100 GHz (28.6% relative bandwidth) and particularly above 45 W (46.5 dBm) in E-band (71–76 GHz, 81–86 GHz) application. This was accomplished by employing a radial off-chip waveguide combiner, which combines 12 broadband GaN MMIC chip modules. The power combining efficiency is 84.5% averaged across the band. In [34], a 100 W W-band GaN PA is also realized by low-loss split-waveguide power combining.

Higher output power can be achieved by on-die, waveguide, and spatial power combining approaches, but the power density remains the intrinsic factor. Ref. [35] demonstrated a wideband W-band GaN amplifier with an unique on-chip traveling-wave combiner design. The minimum outpower is 2 W over the 75–100 GHz bandwidth and reaches a peak of 3 W at 84 GHz, which can be further increased by employing a pulsed mode. A compact package size of 2.75 mm × 5.4 mm gives the amplifier power density of 2.1 W/mm<sup>2</sup>. In [36], a new 80 nm GaN HEMT process is used to fabricate a two-stage cascade unit, each has two transistors with the same gate periphery for high gain matching, which can provide 1.15 W (30.6 dBm) output power and 12.3% PAE at 86 GHz under continuous wave (CW) operation. The 2 × 1.8 mm<sup>2</sup> MMIC reached the power density of 3.6 W/mm at W-band, which is the highest among the published research.

#### 2.4. E-Band Channel Characteristic

#### 2.4.1. Atmospheric Influence

E-band radio transmission could be affected by several atmospheric factors. Among all the factors of regard, the rain rate is tested to have the most decisive influence on E-band radios except for free space loss. In [37], an experimental E-band link has been established to assess impairments of atmospheric attenuation. The link operated over 1.6 km with a constant symbol rate of 3.3 Gbaud at lower E-band. Both the QPSK and 16-QAM have been tested. According to the experiment result, the measured atmospheric attenuation shows good consistency with the ITU model, but the total propagation loss has differences in the absolute values, mainly in rainy conditions. These differences become greater for the cloudy days with cloud percentages over 60%. Moreover, the system shows about 1 dB performance improvement in colder temperatures and a 0.5 dB improvement during the night transmissions (under similar weather conditions). The attenuation caused by water droplets is investigated in [38] based on punctual weather parameters measured over two months. A rain attenuation model has been brought out with specific drop-size distribution (DSD), raindrop velocity, and scattering calculations. The authors also indicated a 0.1 dB negligible atmospheric attenuation difference between the upper cut-off frequency, i.e., 76 GHz, and the lower cut-off frequency, i.e., 71 GHz, at the lower E-band. Yet, the rain attenuation went through a difference of less than 1 dB, which leads to a 2% degradation in EVM at the exploited receiver. Ref. [39] showed the depolarization effect of a 72 GHz beacon signal in tropospheric weather events. The XPD degradation shows to be strongly correlated with rain rate, with variation around 20 dB. However, it behaves differently in the snow event, but it could be concluded that the precipitation type and precipitation rate both underlie the variation of XPD. For the air-to-ground propagation, the ITU (Rec. ITU-R

P.840) indicates that cloud attenuation is related to the product of the attenuation coefficient  $K_l$  ((dB/km)/(g/m<sup>3</sup>)) and liquid water density in the cloud, the former is given by

$$K_{l}(f) = \frac{0.819f}{\varepsilon'' \left(1 + \left(\frac{2 + \varepsilon'}{\varepsilon''}\right)^{2}\right)}$$
(3)

where f (GHz) is the frequency,  $\varepsilon'$  and  $\varepsilon''$  are the real part and imaginary part of the dielectric permittivity of the water content according to the double-Debye model. Evaluation methods are also provided for gases (Rec. ITU-R P.676), scintillation (Rec. ITU-R P.618), and rain (Rec. ITU-R P.618). Though the influence of atmospheric behaviors in E-band millimeter-wave is still at the stage of observing and recording, which needs further investigation and modeling, the ITU model has been verified in various research with mild deviation and is considered an acceptable starting point.

#### 2.4.2. Channel Model

In [40,41], the experimental result showed that the variation in a 60 GHz LoS MIMO channel is relatively slow compared to the typical symbol durations in millimeter wave systems. Minor variations in the amplitude of the impulse responses could be found in a 1.25 GBaud transmission experiment. The mean value of the short-term phase variation is tested  $1.3 \times 10^{-5} \pi$ /Symbol. Notably, it has been observed that the frequency selectivity due to the frontends has a dominant impact on the effective channel in comparison to the multipath effect.

It was once believed that the LoS MIMO channel failed to conduct multi-stream transmission for the lack of scatters on path, till it was proved that full-rank channel matrix could be achieved by spatially configuring the distribution of antennas, that is, the channel condition is strongly relevant to the deployment of MIMO antennas. For example, in reference [42], authors studied the fundamental characteristics of LoS MIMO channel with E-band uniform linear array (ULA) antennas at both receiver and transmitter ends. For the applications with limited physical size, the analysis showed that the maximum transmission distance to support spatial multiplexing with determined spatial streams and SNR is positively correlated with the product of the aperture size at receiver and transmitter arrays instead of numbers. In fact, the antenna arrangement that yields ideal spatial multiplexing can be determined not only for ULAs, but for a variety of array geometries [43–45]. Ref. [46] modeled the LoS MIMO channel capacity with respect to the arrays-of-subarrays (AOSAs) configuration and showed that quasi-optimal capacity could be reached as well as ULA with proper configuration and with less complexity in RF chains and precoders. The work in [47] considered a paradigmatic scenario, i.e., obliquely deployed ULA for strong LoS MIMO communications. By factorizing the channel gain matrix, the author proposed the orthogonal and rank-deficient arrangement with respect to the Rayleigh distance factor. Meanwhile, asymmetric LoS MIMO topologies with high spatial resolution on one side of the link, is proved capable of providing higher robustness for having full-rank channel matrices with distance and direction variation. This model is more practical for unfixed LoS MIMO communication.

Though the E-band backhaul channel for urban microcell and vehicle scenarios has been extensively investigated and modeled in [48–50], they are not suitable for long-range backhaul transmission over 10 km. If we consider a  $2 \times 2$  MIMO with 10 km cover range at 73.5 GHz, according to the Rayleigh criterion

$$R_{Ray} = \frac{N_r d_t d_r}{\lambda} \tag{4}$$

where  $R_{Ray}$  denotes the Rayleigh distance,  $\lambda$  is the wavelength,  $N_r$  is the number of antennas at the receiver,  $d_t$  and  $d_r$  denote the antenna spacing at the transmitter and the

receiver, respectively. The product of antenna spacing at Tx and Rx would be at least 20 m<sup>2</sup>. For scenarios where antenna spacing is limited, e.g., an aircraft pod, such spacing will be unavailable, much less for a more extended range. Therefore, the deployment methodology for long-range transmission over the Rayleigh distance, as well as other in-depth characterization of E-band LoS MIMO channels still remains a challenging issue in the millimeter-wave realm and needs to be further studied. As a step in this direction, ref. [51] proposed a method of introducing specially designed phase variation, e.g., by adding dielectric material in the transmission path, to improve channel capacity. Different phase shifts can be introduced into different channels of LoS MIMO to reduce their correlation, which is further determined by the shapes and types of the dielectric medium. The extended LoS MIMO channel model with additional phase shift is showed to be capable of reducing the antenna aperture required by LoS MIMO system under the Rayleigh distance criterion or extending the Rayleigh distance at a specific antenna aperture. This could be enlightening for an E-band long-range air-to-ground link modeling in the presence of a single-layer cloud. Conventional cloud modeling, such as Cloud Sense Simulation (CSS) and modified Fractal Brownian Motion (FBM), can give meteorologically and visually feasible models of clouds with great complexity and show poor adaption to the air-toground LoS MIMO scenario. We therefore proposed a cloud modeling method to analyze the additional phase variation obtained by electromagnetic waves in the single-layer cloud with randomly distributed ice and moisture component, which shows good consistency to the result in the practical measurement which is reviewed in Section 3. The model, as well as a preliminary evaluation of the influence of cloud in air-to-ground LoS MIMO channel, is presented in [52].

#### 2.5. Digital Baseband

#### 2.5.1. Modulation Scheme

High-order modulations are preferred in E-band high-speed transmission but are usually limited by their implementation complexity and the high sensitivity to phase noise and IQ imbalance. Thus various automatic modulation classification (AMC) methods have been brought out to identify the modulation scheme adaptively based on features extraction [53,54]. On the other hand, constellation shaping has emerged as an attractive technique to improve the SNR of M-QAM systems in linear regimes [55]. By probabilistically and geometrically arranging the signal mapping, the modulation performance can be customized.

In [56], probabilistic amplitude shaping combined with orthogonal circulant matrix transform (OCT) precoding was proposed and experimentally demonstrated in a 64-QAM orthogonal frequency division multiplexing (OFDM) based W-band radio over fiber (RoF) system. Authors in [57] proposed a new probabilistic shaping scheme for 16 and 64 amplitude phase shift keying (APSK). The source was converted into a stream of Rayleigh distributed symbols in the first quadrant by distribution matcher (DM). Thereafter, a systematic low-density parity check (LDPC) encoder was used to generate 2 parity bits per symbol to determine the quadrant for the transmitted symbol. According to the simulation, the proposed PS-APSK modulation could provide 0.9 to 2.1 dB lower peak to average power ratio (PAPR) and higher SNR compared to PS-QAM at the same average power. In terms of geometric shaping, ref. [58] proposed an optimized 4-4-4-N and C-32QAM-N modulation scheme for the millimeter-wave receivers. The proposed circular 16-QAM (4-4-4-N) is constructed by minimizing the average energy at the determined minimum Euclidean distance (MED) and then tuning the distance between the concentric rings to minimize PAPR. Better tolerance in critical phase can be obtained using the proposed scheme compared to the existing works reported in the literature. The result is further verified in an experimental 60 GHz millimeter wave transmission link. For low-PAPR modulation in energy-efficient applications, research [59] proposed constrained phaseshift keying (CPSK) in the discrete Fourier transformation (DFT) spread OFDM system. Additional constraints are applied to conventional PSK, where the symbol transitions are

limited to some of the closest neighbors of the current symbol. The bit-to-symbol mapping works in even and odd time mode with the presence of a constellation rotation to further reduce the PAPR. The authors later evaluated the modulation by applying the PA model defined for IEEE 802.11ad in [60]. While attaining a similar EVM performance, the CPSK has a 1-2dB cut down in PAPR and acts more positively in maximizing the PA output power compared to the conventional QPSK and  $\pi/4$ -QPSK with no extra complexity introduced to the modulator, which seems appealing in the energy-constrained scenario.

#### 2.5.2. LoS MIMO Equalization

Conventional equalization methods, i.e., zero-forcing (ZF) and minimum-mean-squareerror (MMSE), are commonly used in MIMO systems. With development in machine learning, multiple artificial intelligence (AI) strategies are also deployed in equalization to achieve faster execution at the signal processing level with highly-parallel computing [61]. Considering the characteristic of LoS MIMO systems, plenty of research works have been brought out to improve the equalization performance. Authors in [62] proposed a temporal efficient frame structure for stationary LoS MIMO systems with frequency-selective and time-varying components. Orthogonal training sequences were sent as preambles and midambles to estimate the channel state information (CSI) and carrier frequency offset (CFO), which are further introduced in IAI suppression. According to the decomposition of the effective channel matrix, the IAI was suppressed successively by RX CFO de-rotation, MIMO equalization and TX CFO de-rotation with linear complexity. The frequency-selectivity is found rather similar for different streams due to the usage of the similar analog frontend (AFE), so the authors suppressed ISI in parallel by exploiting a fixed MMSE-finite impulse response (FIR) filter per stream, as shown in Figure 4. The proposed low-complexity algorithm was verified in a 60 GHz hardware-in-the-loop  $2 \times 2$  LoS MIMO demonstrator and outperformed the joint equalization using the general version decision-directed least-mean-square (DD-LMS) algorithm.



Figure 4. Block diagram of the proposed 'STIAI+PISIE' structure in [62].

A sequential channel equalization was proposed in [63] to reduce the complexity and improve energy efficiency. The LoS MIMO matrix was factorized into three parts, with the diagonal matrices outside and the middle one dominated by an inverse-DFT matrix considering arrangement displacements along the transmission direction. Equalization was performed sequentially in reverse order with respect to the channel factorization, where DFT operations can exploit the benefits of the butterfly designs in fast Fourier transform (FFT) to reduce the complexity. The overall complexity of the proposed equalization is almost linearly with respect to the multiplexing streams. A 60 GHz  $2 \times 2$  LoS MIMO link was used to verify the proposed equalization, which has approximate performance in comparison to the ZF algorithm with lower complexity. In [64], a novel adaptive space-time equalization for the frequency-selective channel has been proposed to handle geometric misalignment for LoS MIMO, which could introduce additional ISI and crossstream interference (CSI). A Spatial oversampling was conducted by increasing the number of receivers. Adaptive windows in the time domain were employed at each receiver to capture a certain energy level of the currently desired symbol affected by both ISI and CSI to construct a signal space against the interference. The linear minimum mean squared error (LMMSE) equalizer with respect to the temporally and spatially extended channel matrix is then calculated separately for each multiplexed stream to conduct a space-time equalization. The BER performance is simulated better as the spatial oversampling factor and the window length increase, which can be attained adaptively using a greedy strategy. However, the proposed equalization requires a dedicated RF chain for each antenna in the presence of spatial oversampling, which causes increase in cost.

2.5.3. Impairment Compensation

Phase Noise

The phase noise could cause a series of problems, such as constellation rotation, adjacent channel interference (ACI), and inter-carrier interference (ICI) in E-band transceivers if not compensated. The IEEE 802.11 ad [65] recommends the single-pole/zero model to capture the power spectral density (PSD) with

$$S(f) = K_0 \frac{1 + (f/f_z)^2}{1 + (f/f_p)^2}$$
(5)

where  $K_0$  denotes the PN level and  $f_p$  and  $f_z$  are pole and zero frequencies, respectively. A detailed performance evaluation for LoS MIMO systems with the presence of phase noise is performed in [66]. In order to mitigate the impairment caused by phase noise in the millimeter-wave band, a compensation method based on Golay sequence was introduced in [67]. Authors in [68] proposed an iterative decision-aided phase noise compensation scheme for millimeter systems compliant with IEEE 802.11ad standard. The received data block is divided into subblocks where the phase noise is considered stationary. After the equalization, an initial decision was made by considering the phase noise in the whole data block as constant, and by using the former demodulated data, a maximum likelihood (ML) estimation for the phase noise is conducted in each subblock. As the iteration continues, the data block is divided into more pieces, and better data decisions can then be obtained. The proposed algorithm has low complexity and performs better when the iteration number goes greater compared to the conventional decision-directed compensation schemes in [69]. Ref. [70] described a simple phase noise compensation scheme for wideband processing in SC-FDE systems considering the multi-pole/zero PN model, where the recovered unique word (UW) samples are linearly combined to average the error before estimating the phase noise in each data block. The algorithm has the merit of being delay-free and is easy to implement in the physical layer since no iteration is required. Research [71] has developed a low-complexity phase noise suppression for millimeter-wave OFDM systems. In this work, the phase noise is modeled by a specific order of dominant components in the frequency domain. Then integral estimation of the PN-affected channel is conducted by an LMMSE estimator, where a look-up table based on uniformly quantized SNR is used to eliminate the matrix multiplication and inversion. The proposed method has low hardware complexity and has been verified on a low-cost very large scale integrated (VLSI) circuites architecture with 258 K gate count and power consumption of 19.3 mW at 250 MHz clock rate.

IQ Imbalance

The IQ imbalance is usually classified into frequency-independent (FI) imbalance and frequency-dependent (FD) imbalance. The former is introduced by the non-ideality oscillators and shows stationary feature over the bandwidth, while the latter is caused by the frequency-selective components, e.g., matched filters, analog to digital converter (ADCs) and varies over the bandwidth.

A training-based compensation scheme of IQ Imbalance for systems compliant with IEEE 802.11ad standard is researched in [72]. In the SC-FDE scene, the received baseband signal was modeled with respect to the imbalance and channel information. A likelihood

function can be derived with the help of the training sequence. Then the IQ imbalance and channel parameters can be estimated jointly and separately by iteratively maximizing the likelihood function. Finally, compensation and equalization are conducted in the frequency domain. The IQ imbalance on the transmitter side was especially considered in [73] by applying the same model, and the simulation shows that a single iteration is sufficient to approach the benchmark of the ideal case without TX IQ imbalance. Ref. [74] indicated a special frequency-dependent IQ imbalance introduced by AD/DAs, i.e., the in-phase and quadrature timing mismatch (IQTM), which adversely impacts MIMO OFDM-based millimeter-wave system. Various RF distortions, including CFO, PN, sampling frequency offset (SFO), and sampling time offset (STO), were considered in the modeling. The author also proposed a pilot-based joint estimation metric for TX and RX IQTM and the corresponding compensation scheme. Authors in [75] proposed a method to compensate for the frequency-dependent IQ imbalance in wideband direct-conversion millimeter wave transmitters. A transmitter-observation receiver (TOR) design is implemented where a branch of the signal is down-converted to the intermediate frequency (IF) and under-sampled by an ADC after the PA amplification to fit the training algorithm of digital predistortion and IQ compensation. With this TOR feedback procedure, nonideal parameters of the IQ modulator can be estimated by recursive least-squares (RLS) algorithm and compensated by the inverse L-tap structure, as shown in Figure 5. The presented method was experimentally verified in conjunction with the digital predistortion (DPD) process, the compensation scheme shows about 4% reduction in the root normalized mean-square error (RNMSE) of the IQ modulator output signal. The digital predistortion training can be conducted using the same TOR architecture with no restrictions on the IF setting.



**Figure 5.** The baseband-equivalent L-tap model of the I/Q modulator (**left**) and I/Q compensator (**right**).

## PA Distortion

With the application of high-order modulation, the PA linearity grows into one of the major concerns in the E-band long-range backhaul. Previous architectures such as Doherty and envelope-tracking (ET) have been proved to introduce additional distortion into the signal [76]. In this regard, DPD is widely applied in the digital baseband to maximize the output power and energy efficiency.

Ref. [77] first demonstrated DPD improvement in PA efficiency and in-band distortion at E-band with 2 GHz signal bandwidth. A commercial E-band GaAs PA was deployed in a 1.6 Gbaud direct learning loop, the corresponding digital predistorter was implemented by a simple 5th-order memoryless polynomial function. To find the optimal coefficients, the authors started with a behavior modeling technique and optimized the result by exhaustive searching and The Powell's conjugate direction optimization method. The latter is proved to reach the optimal sets within tens of seconds. The proposed DPD can generally increase the output power by 2–3 dB at a specific EVM, which brings the utmost 100% efficiency improvement for a linear PA operation. In [78], an open-loop predistortion was implemented in an E-band communication link to improve signal quality and adjacent channel power (ACP). The authors demonstrated the performance improvement under different channel bandwidths up to 1 GHz from 71 to 76 GHz. By introducing the dynamic deviation reduction (DDR) technique, the Volterra series is simplified and exploited in the DPD. In the performed experimental loop, 3.8 dB of EVM improvement can be observed around the 1 dB compression point after applying the DPD algorithm. The authors also noted that the EVM improvement increases for higher bandwidths, whereas the ACP improvement decreases. In [79], a novel DPD technique that can simultaneously eliminate the cross-polarization interference (XPI) of dual linearly-polarized (DLP) antenna and distortion of PAs in millimeter-wave transmitters was proposed. Firstly, the signal affected by both effects is collected by over-the-air (OTA) tests, then coefficients of both compensation modules can be extracted from the collected signals by applying an iterative decomposition algorithm. Thereafter, predistortion is implemented in the cascaded XPI cancellation and nonlinear compensation. According to the dual-polarized transmission experiment, the proposed DPD had significant improvement in both normalized mean square error (NMSE) and adjacent channel power ratio (ACPR) compared to the memory polynomial based method without XPI compensation.

#### 3. Demonstration of An Experimental Air-to-Ground E-Band Link

To acquire a more profound knowledge of the atmospheric transmission characteristic and multipath effect for E-band long-range air-to-ground channels, we conducted a test campaign at Qinghai, China, located at coordinates 37.7° N, 95.3° E, on 27 September 2021.

#### 3.1. System Overview

As shown in Figure 6, a simplex air-to-ground transmission is implemented. The transmitter was placed in a slow-moving helium balloon platform with a peak rate of 10 m/s, while the receiver was previously arranged on the ground during the flight path. After the ascentstage, the helium balloon platform maintained the level flight at an altitude above 23 km, with the ambient temperature ranging between  $13^{\circ}$  and  $35^{\circ}$  and air pressures between 2.3 and 5.1 kPa. Our transmission experiment was conducted in the predetermined flight path where the cloud gradually receded.



Figure 6. Schematic diagram of the experiment system.

## 3.1.1. Transmitter and Air Payload

The transmitter hardware consists of a probe antenna, a solid-state power amplifier, an upconverter, and baseband field programmable gate array (FPGA) electronics. The beam width of the probe antenna is 60°, which can provide a reachable circular area with a diameter of 23 km on the ground. A GaN power amplifier is adopted with an available output power of 39 dBm at an average PAE of 16%. The FPGA is programmed to continually deliver the data frames previously stored in the ROM with a baud rate of 250 MHz. The baseband signal is first modulated to 800 MHz IF and then upconverted to 74 GHz. The whole transmitter has an EIRP of 44 dBm and is fixed securely in the pod of the helium balloon. Figure 7a shows the setup of the air payload, the upconverter, the local oscillator (LO) generator, and all digital baseband electronics are packaged in the cabinet.



(a) Setup of the air payload (b) Setup of the ground terminal

Figure 7. Implementation of the transmitter over the air and receiver on the ground terminal.

## 3.1.2. Receiver and Ground Terminal

As is shown in Figure 7b, the implementation of the receiver consists of a dualpolarized steerable Cassegrain antenna, an IF module, and digital baseband electronics. The 35 cm Cassegrain antenna can provide 44 dBi gain, resulting in a highly directional beam with HPBW of  $0.7^{\circ}$ . The antenna positioner is functioned by a gimbal motor which is programmed to continually tracks the balloon with the help of GPS information, Figure 8 shows the tracking procedure in the ascent stage of the balloon. After the antenna, a lownoise receiver down-converts the received signal to 800 MHz IF using the ×6 multiplication frequency of the 12.2 GHz LO signal. After being converted to the analog baseband, the signal is digitized and further processed in the digital domain. The processed data are packetized and sent to the software and finally displayed on the monitor. Table 1 illustrates the key parameters of the system.



Figure 8. The steerable Cassegrain antenna tracks the air balloon in the ascent stage.

Parameter	Specification	
Center frequency	74 GHz	
Symbol rate	250 MBaud	
Roll-off factor	0.2	
Modulation scheme	BPSK, QPSK, 16-QAM	
Transmitter		
PA output	38 dBm	
Antenna gain	6 dBi	
EIRP	44 dBm	
Receiver		
Antenna gain	44 dBi	
Antenna polarization	Vertical and horizontal	
Antenna cross-polarization isolation	$\geq$ 30 dB	
Thermal noise	-89 dBm	
Noise figure	7 dB	

Table 1. Experimental air-to-ground link parameters.

The auxiliary equipment at receiver side consists of an AD converter, a disk array and a monitor computer. The 1 Gsps sampling AD is connected directly to the modem for a consecutive data acquisition about 20 min. Sampled signals are then stored into the disk array for future analysis, the computer is used to monitor the beam alignment situation and communication performance of the received signal, including spectrum, constellation and throughput.

#### 3.2. Experimental Result and Discussion

The measurement is conducted under a clear sky condition after the cloud receded, which results in an average atmospheric attenuation of 0.12 dB/km. Real-time DSP including polarization selection combining, synchronization, CFO, and fractional frequency domain equalization is finished in the baseband FPGA with 24.8 W power consumption. Off-line data are also preserved for the future modification in DSP algorithms and the air-to-ground channel analysis, which is further discussed in [52]. Figure 9 shows the 250 MBaud signal quality at LoS distance of 20 km. The transmission remained stable for more than 30 min during the level flight stage. We integrate the Golay sequences into the data frames to estimate the long-range air-to-ground channel, and one of the normalized channel responses averaged over 3 data frames (0.73 ms) at 20 km LoS distance is shown in Figure 10. Due to the conservative hardware setup, there is still room for improvement in the future considering the implementation of MIMO structure.



Figure 9. Constellation diagram of the received siganl and LoS distance of 20 km.



Figure 10. Estimated channel response at LoS distance of 20 km.

#### 4. Open Issues and Future Research Directions

This section discusses a few open research problems that have not been fully studied in the literature and require further research attention.

#### 4.1. Multi-Stream Signal Processing of LoS MIMO Channel over Rayleigh Distance

In a conventional LoS MIMO scenario, the multiplexing gain tends to decrease for longer distances, which limits the transmission rate with a restricted power level. Approaches to reach beyond the Rayleigh distance are still one of the open issues in the long-range wireless backhaul. In this regard, the multiplexing gain variation caused by the heterogeneous medium in the transmission environment still needs to be further studied, since it is possible to extend the coverage by introducing different phase variations in the link [51]. Under this assumption, the efficient MIMO multiplexing scheme, including the optimal antenna deployment for non-uniform antenna array and combination with dual/multi-polarization, is likely to boost the performance of E-band high-speed backhauls.

## 4.2. Atmospheric Influence and Adaptive Link Management

The attenuation effect due to precipitation has been investigated in multiple research. However, the in-depth characteristic of the atmospheric influence, e.g., the phase imbalance and XPD degradation due to different weather conditions [39], as well as the tropospheric turbulence and scintillation effect in E-band long-range channel still need to be further studied. To date the progress is still in preliminary observation. Due to the link variation, an adaptive link management scheme, i.e., the adaptive combination of multiple modulation formats and coding, shall be further considered for a robust and reliable transmission.

#### 4.3. High Accuracy Long-Range Beam Alignment

Due to the highly directional beam in E-band long-range communication, systems are susceptible to beam mismatch. Despite the mechanically assisted reflectors are widely used, dull beam switching and troublesome tracking procedure make it unsuitable for dynamic link and multi-user applications. In this regard, the PAFR, which can offer fast inertia-free scanning with considerable gain and modest cost, might be a feasible solution for future implementation. Resolution to the common defect of PAFR, such as restricted illuminated region and wide scanning deviation, is of interest in recent investigation. On the other hand, the research on joint control algorithms of mechanical and electrical beam steering and fast beam detection method is also essential to dynamic networking.

#### 4.4. High-Speed Multi-Stream Detection and Decoding

Due to the extremely high throughput in future E-band communication, a feasible VLSI implementation is necessary to handle multi-stream signal processing on a large computational scale with up to 100 Gbps. Stochastic computing-based FIR filters and FFT converters [80,81] are considered hopeful to further improve the area and energy efficiency. On the other hand, emerging analog computing brings more significant advantages in speed and power consumption, e.g., [82] have introduced 16 and 64-point analog fast Fourier transform (AFFT) while consuming tens of mW. Resistive processing units can compute matrix-vector multiplication in the analog domain in O(1) time, which gives a considerable boost in signal processing as well as AI and shows the potential to even entirely eliminate the intermediate AD/DA process [83,84].

#### 5. Conclusions

In this paper, we introduce a brief survey of the prevailing long-range, high-capacity Eband communication systems and their enabling technology, i.e., antenna, power amplifier, channel, multiplexing, and digital baseband processing. We experimentally demonstrate an E-band long-range air-to-ground transmission link operated at 74 GHz from a helium balloon to the ground station with a vertical dimension of 20 km, which could be a very start towards the 100 km-100 Gbps objective. In this experimental transmission, we preliminarily verify the system feasibility and channel performance above the troposphere with a conservative hardware setup. Finally, We discuss the open research issues and challenges that deserve future research attention. It is reasonable to be positive that E-band high-speed long haul will thrive in the near future.

Author Contributions: Conceptualization, Z.Z. and N.W.; methodology, N.W.; validation, Z.Z., N.W., X.Z. and Q.L.; formal analysis, Z.Z.; investigation, Z.Z.; resources, N.W. and X.Z.; writing—original draft preparation, Z.Z. and Q.L.; writing—review and editing, N.W., X.Z. and Z.Z.; visualization, Z.Z. and Q.L.; supervision, N.W. All authors have read and agreed to the published version of the manuscript.

**Funding:** This research was funded by the National Key Research and Development Program of China (No. 2020YFB1805001), the National Natural Science Foundation of China (NSFC) under Grant 61871070, 91938202, and the Sichuan Science and Technology Program under Grant 2022YFG0005.

Institutional Review Board Statement: Not applicable.

Data Availability Statement: Not applicable.

Conflicts of Interest: The authors declare no conflict of interest.

## Abbreviations

The following abbreviations are used in this manuscript:

PA	Power amplifier
DARPA	Defense Advanced Research Projects Agency
RF	Radio frequency
LoS	Line of sight
DSP	Digital signal processing
MIMO	Multiple input multiple output
QPSK	Quadrature Phase Shift Keying
QAM	Quadrature amplitude modulation
EIRP	equivalent isotropic radiated power
BER	Bit error rate
AWGN	Additive white Gaussian noise
SNR	Signal to noise ratio
OMT	Orthomode transducer
SSPA	Solid state power amplifier
ADE	Axially displaced ellipse
PAT	Pointing, acquisition and tracking
FoV	Field of view
HPBW	Half power beam width
ILA	Integrated lens antenna
LCP	Liquid crystal polymer
XPD	Cross-polarization discrimination
LHCP	Left hand circular polarization
RHCP	Right hand circular polarization
DoA	Direction of arrival
PLL	Phase lock loop
LNA	Low noise amplifier
IC	Integrated circuit
РСВ	Printed circuit board
RFIC	Radio frequency integrated circuit
EVM	Error vector magnitude
LTCC	Low temperature co-fired ceramic
PAFR	Phased array fed reflector
WAEFR	Wide Angle ESA Fed Reflector
ESA	Electronically scanned array
PAE	Power-added efficiency
MMIC	Monolithic microwave integrated circuit
SC-FDE	Single carrier-frequency domain equalization
SWaP	Small size, weight and power
CW	Continuous Wave
ITU	International Telecommunication Union
DSD	Drop-size distribution
ULA	Uniform linear array
AOSAs	Arrays-of-subarrays
CSS	Cloud Sense Simulation
FBM	Fractal Brownian Motion
AMC	Automatic modulation classification
OCT	Orthogonal circulant matrix transform
OFDM	Orthogonal frequency division multiplexing
RoF	Radio over fiber
APSK	Amplitude phase shift keying

DM	Distribution matcher
LDPC	Low density parity check
PAPR	Peak to average power ratio
MED	Minimum Euclidean distance
CPSK	Constrained phase-shift keying
DFT	Discrete Fourier transformation
ZF	Zero-forcing
MMSE	Minimum-mean-square-error
AI	Artificial intelligence
CSI	Channel state information
CFO	Carrier frequency offset
FIR	Finite impulse response
DD-LMS	Decision-directed least-mean-square
FFT	Fast Fourier transform
CSI	Cross-stream interference
LMMSE	Linear minimum mean squared error
PSD	Power spectral density
ML	Maximum likelihood
UW	Unique word
VLSI	Very large scale integrated circuites
AD	Analog to digital
DA	Digital to analog
ADC	Analog to digital converter
FI	Frequency-independent
FD	Frequency-dependent
IQTM	In-phase and quadrature timing mismatch
SFO	Sampling frequency offset
STO	Sampling time offset
TOR	Transmitter-observation receiver
RLS	Recursive least-squares
RNMSE	Root normalized mean-square error
IF	Intermediate frequency
ET	Envelope-tracking
DPD	Digital predistortion
ACP	Adjacent channel power
DDR	Dynamic deviation reduction
XPI	Cross-polarization interference
DLP	Dual linearly-polarized
OTA	Over-the-air
NMSE	Normalized mean square error
ACPR	Adjacent channel power ratio
FPGA	Field programmable gate array
LO	Local oscillator
GPS	Global Position System
Gsps	Giga sample per second
BPSK	Binary phase shift keying
AFFT	Analog fast Fourier transform

## References

- 1. Uwaechia, A.N.; Mahyuddin, N.M. A comprehensive survey on millimeter wave communications for fifth-generation wireless networks: Feasibility and challenges. *IEEE Access* 2020, *8*, 62367–62414. [CrossRef]
- Woodward, T. What To Do When There's No Fiber: The DARPA 100Gb/s RF Backbone Program. In Proceedings of the Optical Fiber Communication Conference, Los Angeles, CA, USA, 19–23 March 2017; Optica Publishing Group: Hong Kong, 2017; p. Tu3E-6.
- 3. Li, X.; Xiao, J.; Yu, J. Long-distance wireless mm-wave signal delivery at W-band. J. Light. Technol. 2016, 34, 661–668. [CrossRef]
- Harati, P.; Rosello, E.; Dan, I.; Bammidi, E.; Eisenbeis, J.; Tessmann, A.; Schwantuschke, D.; Henneberger, R.; Kallfass, I. E-band downlink wireless data transmission for future satellite communication. In Proceedings of the 2017 Topical Workshop on Internet of Space (TWIOS), Phoenix, AZ, USA, 15–18 January 2017; pp. 1–4.

- Kallfass, I.; Antes, J.; Tessmann, A.; Zwick, T.; Henneberger, R. Multi-Gigabit high-range fixed wireless links at high millimeterwave carrier frequencies. In Proceedings of the 2017 IEEE Radio and Wireless Symposium (RWS), Phoenix, AZ, USA, 15–18 January 2017; pp. 45–48.
- Brown, K.; Brown, A.; Feenstra, T.; Gritters, D.; Ko, E.; O'Connor, S.; Sotelo, M.; Raytheon. Long-range wireless link with fiber-equivalent data rate. In Proceedings of the 2017 IEEE MTT-S International Microwave Symposium (IMS), Honolulu, HI, USA, 4–9 June 2017; pp. 809–811.
- Czegledi, C.B.; Hörberg, M.; Sjödin, M.; Ligander, P.; Hansryd, J.; Sandberg, J.; Gustavsson, J.; Sjöberg, D.; Polydorou, D.; Siomos, D. Demonstrating 139 Gbps and 55.6 bps/Hz spectrum efficiency using 8× 8 MIMO over a 1.5-km link at 73.5 GHz. In Proceedings of the 2020 IEEE/MTT-S International Microwave Symposium (IMS), Los Angeles, CA, USA, 4–6 August 2020; pp. 539–542.
- Kallfass, I.; Henneberger, R.; Sommer, R.; Harati, P.; Dilek, S.; Schoch, B.; Eisenbeis, J.; Marahrens, S.; Palm, S.; Tessmann, A. High System Gain E-Band Link in a Wideband Aircraft-to-Ground Data Transmission. In Proceedings of the 2019 IEEE International Conference on Microwaves, Antennas, Communications and Electronic Systems (COMCAS), Tel-Aviv, Israel, 4–6 November 2019; pp. 1–5.
- Tiwari, A.; del Portillo, I.; Reed, M.; Zhou, H.; Shmueli, D.; Ristroph, G.; Cashion, S.; Zhang, D.; Stewart, J.; Bondalapati, P.; et al. Demonstration of a 40Gbps Bi-directional Air-to-Ground Millimeter Wave Communication Link. In Proceedings of the 2019 IEEE MTT-S International Microwave Symposium (IMS), Boston, MA, USA, 2–7 June 2019; pp. 746–749.
- Rzasa, J.; Ertem, M.C.; Davis, C.C. Pointing, acquisition, and tracking considerations for mobile directional wireless communications systems. In Proceedings of the Laser Communication and Propagation through the Atmosphere and Oceans II, San Diego, CA, USA, 26–27 August 2013; Volume 8874, pp. 47–56.
- Roy, S.S.; Sekhar, T.N. High Efficiency Compact Ground Station Antenna Using Displaced Ellipse Reflector. In Proceedings of the 2011 IEEE Applied Electromagnetics Conference (AEMC), Kolkata, India, 18–22 December 2011; pp. 1–4.
- Jiang, H.; Yao, Y.; Cheng, X.; Xiu, T.; Yu, J.; Chen, X. Axial Displaced Ellipse Reflector Antenna with Adjustable Beamwidth for E-band Wireless Link. In Proceedings of the 2020 Cross Strait Radio Science & Wireless Technology Conference (CSRSWTC), Fuzhou, China, 13–16 December 2020; pp. 1–3.
- Ala-Laurinaho, J.; Aurinsalo, J.; Karttunen, A.; Kaunisto, M.; Lamminen, A.; Nurmiharju, J.; Räisänen, A.V.; Säily, J.; Wainio, P. 2-D Beam-Steerable Integrated Lens Antenna System for 5G \$E\$ -Band Access and Backhaul. *IEEE Trans. Microw. Theory Tech.* 2016, 64, 2244–2255. [CrossRef]
- 14. Shu, C.; Hu, S.; Yao, Y.; Chen, X. A High-Gain Antenna with Dual Circular Polarization for W-band mmWave Wireless Communications. In Proceedings of the 2019 Computing, Communications and IoT Applications (ComComAp), Shenzhen, China, 26–28 October 2019; pp. 29–32.
- Tan, W.; Luo, H.; Zhao, G. Design of a W-band Dual-polarization Monopulse Reflector Antenna. In Proceedings of the 2018 International Applied Computational Electromagnetics Society Symposium-China (ACES), Beijing, China, 29 July–1 August 2018; pp. 1–2.
- 16. Shahramian, S.; Holyoak, M.J.; Baeyens, Y. A 16-Element W-Band Phased-Array Transceiver Chipset With Flip-Chip PCB Integrated Antennas for Multi-Gigabit Wireless Data Links. *IEEE Trans. Microw. Theory Tech.* **2018**, *66*, 14. [CrossRef]
- 17. Shahramian, S.; Holyoak, M.J.; Singh, A.; Baeyens, Y. A Fully Integrated 384-Element, 16-Tile, \$W\$ -Band Phased Array With Self-Alignment and Self-Test. *IEEE J.-Solid-State Circuits* **2019**, *54*, 2419–2434. [CrossRef]
- Repeta, M.; Zhai, W.; Ross, T.; Ansari, K.; Tiller, S.; Pothula, H.K.; Wessel, D.; Li, X.; Cai, H.; Liang, D.; et al. A Scalable 256-Element E-Band Phased-Array Transceiver for Broadband Communications. In Proceedings of the 2020 IEEE/MTT-S International Microwave Symposium (IMS), Los Angeles, CA, USA, 4–6 August 2020; pp. 833–836.
- Chengjin, J.; Jun, W.; Bo, P.; Yan, Z.; Heyminck, S. PAF application on a large spherical reflector to survey 100 times faster than the FAST. In Proceedings of the 2021 IEEE International Symposium on Antennas and Propagation and USNC-URSI Radio Science Meeting (APS/URSI), Singapore, 4–10 December 2021; pp. 45–46.
- Navarrini, A.; Monari, J.; Scalambra, A.; Melis, A.; Concu, R.; Naldi, G.; Maccaferri, A.; Cattani, A.; Ortu, P.; Roda, J.; et al. Design of PHAROS2 Phased Array Feed. In Proceedings of the 2018 2nd URSI Atlantic Radio Science Meeting (AT-RASC), Gran Canaria, Spain, 28 May–1 June 2018; pp. 1–4.
- Hovey, G.; Burgess, T.; Carlson, B.; Ljusic, Z.; Veidt, B.; Zhang, H. A broadband FPGA digital beamformer for the advanced focal array demonstrator (AFAD). In Proceedings of the 2011 XXXth URSI General Assembly and Scientific Symposium, Istanbul, Turkey, 13–20 August 2011; pp. 1–4.
- Chippendale, A.; Beresford, R.; Deng, X.; Leach, M.; Reynolds, J.; Kramer, M.; Tzioumis, T. Testing a modified ASKAP Mark II
  phased array feed on the 64 m Parkes radio telescope. In Proceedings of the 2016 International Conference on Electromagnetics in
  Advanced Applications (ICEAA), Cairns, QLD, Australia, 19–23 September 2016; pp. 909–912.
- Cooley, M. Phased Array Fed Reflector (PAFR) Antenna Architectures for Space-Based Sensors. In Proceedings of the 2015 IEEE Aerospace Conference, Big Sky, MT, USA, 7–14 March 2015; pp. 1–11.
- Al-Rawi, A.; Dubok, A.; Geluk, S.J.; de Hon, B.P.; Herben, M.H.A.J.; Smolders, A.B. Increasing the EIRP by Using FPA-fed Reflector Antennas. In Proceedings of the 2016 IEEE International Symposium on Antennas and Propagation (APSURSI), Fajardo, PR, USA, 26 June–1 July 2016; pp. 1623–1624.

- Dubok, A.; Al-Rawi, A.; Herben, M.H.A.J.; Gerini, G.; Smolders, A.B. Wide-Angle Scanning Reflector Configuration for Focal Plane Arrays. In Proceedings of the 2016 IEEE International Symposium on Antennas and Propagation (APSURSI), Fajardo, PR, USA, 26 June–1 July 2016; pp. 1625–1626.
- 26. Dubok, A.; Gerini, G.; Smolders, A.B. Extreme Scanning Double Shaped-Reflector Antenna With Multiple Interactions for Focal Plane Array Applications. *IEEE Trans. Antennas Propag.* 2020, *68*, 5686–5690. [CrossRef]
- Hand, T.H.; Torres, J.M.; Nielsen, T.; Gustafson, J.D.; Moschetti, P.D. A Millimeter-Wave Wide Band, Wide Scanning Phased Array-Fed Reflector Architecture. In Proceedings of the 2021 IEEE International Symposium on Antennas and Propagation and USNC-URSI Radio Science Meeting (APS/URSI), Singapore, 4–10 December 2021; pp. 1407–1408.
- Camarchia, V.; Quaglia, R.; Piacibello, A.; Nguyen, D.P.; Wang, H.; Pham, A.V. A Review of Technologies and Design Techniques of Millimeter-Wave Power Amplifiers. *IEEE Trans. Microw. Theory Tech.* 2020, 68, 2957–2983. [CrossRef]
- Wang, H.; Wang, F.; Nguyen, H.T.; Li, S.; Huang, T.Y.; Ahmed, A.S.; Smith, M.E.D.; Mannem, N.S.; Lee, J. Power amplifiers performance survey 2000-present. *Ga. Tech Electron.-Micro-Syst. Lab (Gems) Tech. Rep.* 2020. Available online: https://gems.ece. gatech.edu/PA\_survey.html (accessed on 29 October 2022).
- Margomenos, A.; Kurdoghlian, A.; Micovic, M.; Shinohara, K.; Brown, D.F.; Corrion, A.L.; Moyer, H.P.; Burnham, S.; Regan, D.C.; Grabar, R.M.; et al. GaN Technology for E, W and G-Band Applications. In Proceedings of the 2014 IEEE Compound Semiconductor Integrated Circuit Symposium (CSICS), La Jolla, CA, USA, 19–22 October 2014; pp. 1–4.
- Xu, P.; Cheng, Z.; Zhang, Z.; Meng, M. High Efficiency GaN HEMT E-Band Power Amplifier MMIC with 1.8W Output Power. In Proceedings of the 2020 IEEE 5th International Conference on Integrated Circuits and Microsystems (ICICM), Nanjing, China, 23–25 October 2020; pp. 130–133.
- 32. Brown, K.W.; Gritters, D.M.; Kazemi, H. W-and G-band solid state power combining. In Proceedings of the 2015 IEEE Compound Semiconductor Integrated Circuit Symposium (CSICS), New Orleans, LA, USA, 11–14 October 2015; pp. 1–4.
- Schellenberg, J.; Tran, A.; Bui, L.; Cuevas, A.; Watkins, E. 37 W, 75-100 GHz GaN Power Amplifier. In Proceedings of the 2016 Ieee Mtt-S International Microwave Symposium (IMS), San Francisco, CA, USA, 22–27 May 2016.
- Soric, J.; Kolias, N.; Saunders, J.; Kotce, J.; Brown, A.; Rodenbeck, C.; Gyurcsik, R. A 100-W W-Band GaN SSPA. IEEE Microw. Wirel. Components Lett. 2022. [CrossRef]
- Schellenberg, J.M. A 2-W W-band GaN traveling-wave amplifier with 25-GHz bandwidth. IEEE Trans. Microw. Theory Tech. 2015, 63, 2833–2840. [CrossRef]
- Niida, Y.; Kamada, Y.; Ohki, T.; Ozaki, S.; Makiyama, K.; Minoura, Y.; Okamoto, N.; Sato, M.; Joshin, K.; Watanabe, K. 3.6 W/Mm High Power Density W-band InAlGaN/GaN HEMT MMIC Power Amplifier. In Proceedings of the 2016 IEEE Topical Conference on Power Amplifiers for Wireless and Radio Applications (PAWR), Austin, TX, USA, 24–27 January 2016; pp. 24–26.
- Manoliu, L.; Henneberger, R.; Tessmann, A.; Seidel, J.; Eppard, M.; Wrana, D.; Schoch, B.; Kallfass, I. Measurements of Atmospheric Attenuation in an Outdoor Wireless E/W-Band Communication Link. In Proceedings of the 2022 14th German Microwave Conference (GeMiC), Ulm, Germany, 16–18 May 2022; pp. 208–211.
- Manoliu, L.; Henneberger, R.; Tessmann, A.; Seidel, J.; Eppard, M.; Kallfass, I. Impairments of Atmospheric Attenuation on a Wideband E-Band Outdoor Communication Link. In Proceedings of the 2021 51st European Microwave Conference (EuMC), London, UK, 4–6 April 2022; pp. 990–993.
- Keeley, J.D.; Lane, S.; Murrell, D.; Hong, E.; Tarasenko, N.; Christodoulou, C.; Gesner, R. Depolarization of a V-band (72 GHz) Circularly Polarized Beacon Signal Due to Tropospheric Weather Events. In Proceedings of the 2020 IEEE International Symposium on Antennas and Propagation and North American Radio Science Meeting, Montreal, QC, Canada, 5–10 July 2020; pp. 1115–1116.
- Halsig, T.; Cvetkovski, D.; Grass, E.; Lankl, B. Statistical Properties and Variations of LOS MIMO Channels at Millimeter Wave Frequencies. In Proceedings of the 2018 22nd International ITG Workshop on Smart Antennas (WSA), Bochum, Germany, 14–16 March 2018; p. 6.
- Halsig, T.; Cvetkovski, D.; Grass, E.; Lankl, B. Measurement results for millimeter wave pure LOS MIMO channels. In Proceedings of the 2017 IEEE Wireless Communications and Networking Conference (WCNC), San Francisco, CA, USA, 19–22 March 2017; pp. 1–6.
- 42. Wang, P.; Li, Y.; Yuan, X.; Song, L.; Vucetic, B. Tens of Gigabits Wireless Communications Over E-Band LoS MIMO Channels with Uniform Linear Antenna Arrays. *IEEE Trans. Wirel. Commun.* **2014**, *13*, 3791–3805. [CrossRef]
- Li, R.; He, S.; Huang, Y.; Li, Y.; Yang, L. Analysis of Panel Antenna Arrays in Los MIMO System. *IEEE Access* 2018, 6, 23303–23315. [CrossRef]
- 44. Jeon, Y.; Gil, G.T.; Lee, Y.H. Design and Analysis of LoS-MIMO Systems With Uniform Circular Arrays. *IEEE Trans. Wirel. Commun.* 2021, 20, 4527–4540. [CrossRef]
- Okada, R.; Nishimori, K. Circular Antenna Arrangement for LOS-MIMO Transmission Independent for the Transmission Distance Using a Genetic Algorithm. In Proceedings of the 2020 International Symposium on Antennas and Propagation (ISAP), Osaka, Japan, 25–28 January 2021; pp. 515–516.
- Do, H.; Lee, N.; Lozano, A. Capacity of Line-of-Sight MIMO Channels. In Proceedings of the 2020 IEEE International Symposium on Information Theory (ISIT), Los Angeles, CA, USA, 21–26 June 2020; pp. 2044–2048.
- 47. Song, X.; Ma, S.; Neuhaus, P.; Wang, W.; Gao, X.; Fettweis, G. On Robust Millimeter Wave Line-of-Sight MIMO Communications with Few-Bit ADCs. *IEEE Trans. Wirel. Commun.* **2022**, 1. [CrossRef]

- Abbas, T.; Qamar, F.; Ahmed, I.; Dimyati, K.; Majed, M.B. Propagation Channel Characterization for 28 and 73 GHz Millimeter-Wave 5G Frequency Band. In Proceedings of the 2017 IEEE 15th Student Conference on Research and Development (SCOReD), Putrajaya, Malaysia, 13–14 December 2017; pp. 297–302.
- Dutty, H.B.H.; Mowla, M.M.; Mou, M.A. Millimeter Wave Channel Modeling for Aviation Backhaul Networks in 5G Communications. In Proceedings of the 2019 22nd International Conference on Computer and Information Technology (ICCIT), Dhaka, Bangladesh, 18–20 December 2019; pp. 1–5.
- Momo, S.H.A.; Mowla, M.M. Statistical Analysis of an Outdoor mmWave Channel Model at 73 GHz for 5G Networks. In Proceedings of the 2019 International Conference on Computer, Communication, Chemical, Materials and Electronic Engineering (IC4ME2), Rajshahi, Bangladesh, 11–12 July 2019; pp. 1–4.
- 51. Hälsig, T.; Lankl, B. Array size reduction for high-rank LoS MIMO ULAs. IEEE Wirel. Commun. Lett. 2015, 4, 649–652. [CrossRef]
- 52. Wei, N.; Tang, S.; Zhang, Z. Preliminary Analysis of Channel Capacity in Air to ground LoS MIMO Communication Based on A Cloud Modeling Method. *arXiv* 2022, arXiv:cs.IT/2210.10648.
- Hamed, H.A.; Abdullah, A.K.; Sadkhan, S.B.; Abdullah, A.A. M-ary Quadrature Ampplitude Modulation Classification Using Skewness and Kurtosis. In Proceedings of the 2021 1st Babylon International Conference on Information Technology and Science (BICITS), Babil, Iraq, 28–29 April 2021; pp. 109–112.
- 54. Meng, F.; Chen, P.; Wu, L.; Wang, X. Automatic modulation classification: A deep learning enabled approach. *IEEE Trans. Veh. Technol.* **2018**, *67*, 10760–10772. [CrossRef]
- 55. Vassilieva, O.; Kim, I.; Palacharla, P. Probabilistic vs. Geometric Constellation Shaping: What are the keys to ultimate optical reach and capacity? In Proceedings of the 2021 IEEE Photonics Conference (IPC), Virtual, 18–21 October 2021; pp. 1–2.
- 56. Wu, K.; He, J.; Zhou, Z.; Shi, J. Probabilistic amplitude shaping for a 64-QAM OFDM W-band RoF system. *IEEE Photonics Technol. Lett.* **2019**, *31*, 1076–1079. [CrossRef]
- Poor, S.Z.; Forouzan, A.R.; Parvaresh, F. A New Probabilistic Constellation Shaping Scheme for APSK Modulation. In Proceedings of the 2021 Iran Workshop on Communication and Information Theory (IWCIT), Tehran, Iran, 19–20 May 2021; pp. 1–4.
- 58. Mvone, R.; Hannachi, C.; Hammou, D.; Moldovan, E.; Tatu, S. Optimization of 16-QAM and 32-QAM Constellations for Mitigating Impairments of Phase Noise in Millimeter-Wave Receivers. *IEEE Trans. Wirel. Commun.* **2021**, *21*, 3605–3616. [CrossRef]
- 59. Nasarre, I.P.; Levanen, T.; Valkama, M. Constrained PSK: Energy-efficient modulation for sub-THz systems. In Proceedings of the 2020 IEEE International Conference on Communications Workshops (ICC Workshops), Dublin, Ireland, 7–11 June 2020; pp. 1–7.
- 60. Perahia, E.; Park, M.; Stacey, R.; Zhang, H.; Yee, J.; Ponnampalam, V.; Erceg, V.; Bourdoux, A.; Cordeiro, C.; Maslennikov, R.; et al. IEEE P802. 11 wireless LANs TGad evaluation methodology. *IEEE* 2010, *802*, 3–5.
- 61. Carrera, D.F.; Vargas-Rosales, C.; Yungaicela-Naula, N.M.; Azpilicueta, L. Comparative study of artificial neural network based channel equalization methods for mmWave communications. *IEEE Access* **2021**, *9*, 41678–41687. [CrossRef]
- 62. Song, X.; Hälsig, T.; Cvetkovski, D.; Rave, W.; Lankl, B.; Grass, E.; Fettweis, G. Design and experimental evaluation of equalization algorithms for line-of-sight spatial multiplexing at 60 GHz. *IEEE J. Sel. Areas Commun.* **2018**, *36*, 2570–2580. [CrossRef]
- 63. Song, X.; Cvetkovski, D.; Rave, W.; Grass, E.; Fettweis, G. Sequential channel equalization in strong line-of-sight MIMO communication. *IEEE Trans. Wirel. Commun.* **2018**, *18*, 340–356. [CrossRef]
- Giridhar, L.; Rasekh, M.E.; Sezer, A.D.; Madhow, U. Adaptive Space-Time Equalization with Spatial Oversampling for Misaligned LoS MIMO. In Proceedings of the 2022 IEEE Wireless Communications and Networking Conference (WCNC), Austin, TX, USA, 10–13 April 2022; pp. 1569–1574.
- 65. The IEEE Standards Association. Wireless LAN Media Access Control (MAC) and Physical Layer (PHY) Specifications. Available online: http://standards.ieee.org/getieee802/ (accessed on 29 October 2022).
- Hälsig, T.; Lankl, B. Performance Evaluation of LOS MIMO Systems under the Influence of Phase Noise. In Proceedings of the WSA 2015; 19th International ITG Workshop on Smart Antennas, VDE, Ilmenau, Germany, 3–5 March 2015; pp. 1–5.
- Yao, T.; Wang, X.; Zhou, H.; Tang, S.; Xu, C. A Method of Phase Noise Estimation and Compensation Based on Golay Sequence in Time Domain for SC-FDE Millimeter-Wave Communication Systems. In Proceedings of the 2020 IEEE 6th International Conference on Computer and Communications (ICCC), Chengdu, China, 11–14 December 2020; pp. 278–282.
- 68. Cheng, X.; Lou, N.; Yuan, B. Iterative decision-aided compensation of phase noise in millimeter-wave SC-FDE systems. *IEEE Commun. Lett.* **2016**, *20*, 1030–1033. [CrossRef]
- 69. Suyama, S.; Onodera, J.; Suzuki, H.; Fukawa, K. Decision-directed phase noise compensation for millimeter-wave single carrier transmission systems with frequency-domain equalization. In Proceedings of the 2009 European Wireless Technology Conference, Rome, Italy, 28–29 September 2009; pp. 9–12.
- Park, H.S. Phase noise compensation method in sub-THz SC-FDE systems. In Proceedings of the 2021 International Conference on Information and Communication Technology Convergence (ICTC), Jeju Island, Republic of Korean, 20–22 October 2021; pp. 1428–1430.
- Chung, M.; Prabhu, H.; Sheikh, F.; Edfors, O.; Liu, L. Low-Complexity Fully-Digital Phase Noise Suppression for Millimeter-Wave Systems. In Proceedings of the 2020 IEEE International Symposium on Circuits and Systems (ISCAS), Virtual, 10–21 October 2020; pp. 1–5.
- 72. Cheng, X.; Luo, Z.; Li, S. Joint estimation for I/Q imbalance and multipath channel in millimeter-wave SC-FDE systems. *IEEE Trans. Veh. Technol.* **2015**, *65*, 6901–6912. [CrossRef]

- Cheng, X.; Luo, Z. Compensation of transmitter I/Q imbalance in millimeter-wave SC-FDE systems. *IEEE Trans. Veh. Technol.* 2016, 66, 4472–4476. [CrossRef]
- Minn, H.; Zhan, Q.; Al-Dhahir, N.; Huang, H. In-phase and quadrature timing mismatch estimation and compensation in millimeter-wave communication systems. *IEEE Trans. Wirel. Commun.* 2017, 16, 4317–4331. [CrossRef]
- Almoneer, M.; Mitran, P.; Boumaiza, S. I/Q imbalance compensation in wideband millimeter-wave transmitters using a single undersampling ADC. *IEEE Trans. Circuits Syst. Regul. Pap.* 2020, 67, 2730–2738. [CrossRef]
- 76. Katz, A.; Wood, J.; Chokola, D. The evolution of PA linearization: From classic feedforward and feedback through analog and digital predistortion. *IEEE Microw. Mag.* 2016, *17*, 32–40. [CrossRef]
- 77. Tang, Q.; Zhou, H.; Tiwari, A.; Stewart, J.; Qu, Q.; Zhang, D.; Hemmati, H. Experimental demonstration of digital pre-distortion for millimeter wave power amplifiers with GHz bandwidth. In Proceedings of the 2018 IEEE Topical Conference on RF/Microwave Power Amplifiers for Radio and Wireless Applications (PAWR), Anaheim, CA, USA, 14–17 January 2018; pp. 58–61.
- Schoch, B.; Wiewel, F.; Wrana, D.; Manoliu, L.; Haussmann, S.; Tessmann, A.; Kallfass, I. Performance Optimization of an E-Band Communication Link using Open-Loop Predistortion. In Proceedings of the 2022 14th German Microwave Conference (GeMiC), Ulm, Germany, 16–18 May 2022; pp. 216–219.
- Wu, Q.; Yin, H.; Zhu, Y.; Chen, P.; Zhu, X.W.; Yu, C.; Hong, W. Digital Predistortion for Dual Linearly-Polarized Millimeter Wave Transmitters with Cross-Polarization Interference. In Proceedings of the 2021 IEEE MTT-S International Wireless Symposium (IWS), Nanjing, China, 23–26 May 2021; pp. 1–3.
- Mahesh, V.; Shahana, T. Design and synthesis of FIR filter banks using area and power efficient Stochastic Computing. In Proceedings of the 2020 Fourth World Conference on Smart Trends in Systems, Security and Sustainability (WorldS4), London, UK, 27–28 July 2020; pp. 662–666.
- Gonzalez-Guerrero, P.; Guo, X.; Stan, M.R. ASC-FFT: Area-efficient low-latency FFT design based on asynchronous stochastic computing. In Proceedings of the 2019 IEEE 10th Latin American Symposium on Circuits & Systems (LASCAS), Armenia, Colombia, 24–27 February 2019; pp. 117–120.
- Li, J.; Chao, X.; Li, Q. 16-and 64-Point Analog Computing of FFT with Improved Performance and Efficiency. In Proceedings of the 2022 17th Conference on Ph. D Research in Microelectronics and Electronics (PRIME), Villasimius, Italy, 12–15 June 2022; pp. 197–200.
- Chen, X.; Song, T.; Han, Y. RRAM-based Analog In-Memory Computing. In Proceedings of the 2021 IEEE/ACM International Symposium on Nanoscale Architectures (NANOARCH), AB, Canada, 8–10 November 2021; pp. 1–6.
- Assaf, H.; Savaria, Y.; Sawan, M. Vector Matrix Multiplication Using Crossbar Arrays: A Comparative Analysis. In Proceedings of the 2018 25th IEEE International Conference on Electronics, Circuits and Systems (ICECS), Bordeaux, France, 9–12 December 2018; pp. 609–612.